Author's response to reviews

Title: Discussing life expectancy with surgical patients: Do patients want to know and how should this information be delivered?

Authors:

Michael G Clarke (mgclarkey@doctors.org.uk)
Katherine P Kennedy (katherinephilippakennedy@hotmail.com)
Ruaraidh P MacDonagh (rumac@doctors.org.uk)

Version: 5 Date: 12 June 2008

Author's response to reviews:

Have made more revisions
References: Numbers in brackets and before full stops
Section headings: Removed unnecessary capital letters
Attached files: Removed this wording from document